

Format for ANSWERING REVIEWERS



November 21, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (Non-alcoholic Fatty Liver Disease: The Diagnosis and Management: 11425-review.doc).

Title: Non-alcoholic Fatty Liver Disease: The Diagnosis and Management.

Author: Shehab M. Abd El- Kader and Eman M. Salah El-Den Ashmawy

Name of Journal: *World Journal of Hepatology*

ESPS Manuscript NO: 11425

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

- In the section of DIAGNOSIS, Laboratory testing should also include smooth muscle antibodies. (Done)
- On page 19, paragraph 3 - first word should be nonalcoholic steatohepatitis (NASH) instead of alcoholic Steatohepatitis (ASH). (Done)
- In the treatment section, dosing of vitamin E is important and should be mentioned. (Done)
- Nonetheless, the authors should insert further details concerning the histological pattern of pediatric NAFLD differs from that in adults. (Done)
- Furthermore, the crucial role of the macrophages in NAFLD/NASH evolution should be underlined. (Done)
- Some typographical and punctuation errors are present in the manuscript. (Done)
- It is now clear that the interplay between oxidative stress and immunity has a critical role in NASH progression thus the following papers should be cited in the pathogenesis session. (Done)
- The review is interesting. However, Language should be improved and the reference list updated (Done)
- More references are needed, some sentences are lacking of their references (Done)
- The quality of English is poor, there are a lot of spelling errors, and the text should be revised (Done)
- There are a lot of repetitions in all the text, a shortening work is needed. (Done)

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Hepatology*

Sincerely yours,

A handwritten signature in blue ink, appearing to be 'Shehab M. Abd El-Kader'.

Prof. Shehab M. Abdel-Kader

Prof. of Physical Therapy, Cairo University, Egypt

Prof. of Physical Therapy, KAU, Saudi Arabia

Accreditation and Quality coordinator, FAMS , KAU.

Cardiopulmonary and Geriatrics Track

Kingdom of Saudi Arabia, Jeddah.

P.O Box: 80324, Jeddah 21589

Cell Phone: +966569849276

Personal Website: <http://Salmuzain.kau.edu.sa>

E-mail: profshehab@live.com